Mauna Loa Sol ar Observatory Cbserver’'s Log

Fri Sep 9 16:40:01 GV 1994
Year: 94 Doy: 252
onserver: yasukawa

WEATHER COMMENT: Fri Sep 9 16:40:23 GVI 1994
Cool, Clear at the nmonment, extensive altocunulus deck just above sun
extends to west horizon. Light south breeze. Tenmp = 46F

Fri Sep 9 16:49:37 GMI 1994: Patrol Start

MKITIT COMMENT: Fri Sep 9 16:49:44 GVII 1994

W ped front surface of OL with optical w pe (dry), and brushed it off.
Looks much better-no streaks.

Fri Sep 9 17:02:01 GMI 1994: Calibration

LOML COWENT: Fri Sep 9 17:05:12 GVII 1994

L00103 was conpl ete and popped out. Renoved it and | oaded L00105
into unit O.

WEATHER COMMENT: Fri Sep 9 17:09:03 GV 1994
In al tocunul us deck.

COMMENT: Fri Sep 9 17:09:49 GMI 1994
Dusted off front filters of sky tx tel escope and spar guider

COMMENT: Fri Sep 9 17:10:34 GMI 1994
St opped MKI I and DPMON whil e awaiting altocurmul us burn-off.

COMMENT: Fri Sep 9 18:40:12 GMI 1994

Al tocumul us starting to break up. Reconfigured done slot even though
MKITl is now partially domed out yet by the bottom shutter segnent.
Brushed front surface of MKIII OL.

Fri Sep 9 18:46:44 GVI 1994: Patrol Start

COWMENT: Fri Sep 9 18:50:21 GVII 1994
MKI'I'l and DPMON observations resuming. Still in broken altocunul us.

Fri Sep 9 19:02:00 GMI 1994: Calibration

WEATHER COMVENT: Fri Sep 9 19:19:42 GMI 1994



Al tocumul us now only scattered. May start getting conplete coronal
scans.

WEATHER COMMENT: Fri Sep 9 19:30:19 GMI 1994
Al tostratus form ng over head.

WEATHER COMMENT: Fri Sep 9 19:43:05 GVI 1994
d ear

Fri Sep 9 20:02:02 GV 1994: Calibration

MKI Il COWMMENT: Fri Sep 9 20:07:24 GMVI 1994
Fi ni shed calibrating channel 1. Skipping dark scan this tinme unti
channnel 0 calibration.

MKITIT COMMENT: Fri Sep 9 20:10:34 GVII 1994
Cal i brating channel 0

DPMON COMMENT: Fri Sep 9 20:19:57 GMI 1994
Reconfigured to disc preferred node.

WEATHER COMMENT: Fri Sep 9 20:28:50 GMI 1994
Sky getting bright and it has been quite noisy.

WEATHER COMMENT: Fri Sep 9 20:41:10 GMVI 1994
cirrostratus form ng overheaad.

DPMON COMMENT: Fri Sep 9 20:43:37 GMI 1994
Abundant tape 1/O error nessages dunped to consol e screen

Fri Sep 9 20:47:00 GMI 1994: Patrol End

DPMON COMMENT: Fri Sep 9 20:53:47 GMI' 1994

Went into manual node and "pressed"” EJECT button, renoved P00156.
Cl eaned heads with cl eaning cassette, |oaded P00157.

Restarting Patrol.

Fri Sep 9 20:56:12 GMII 1994: Patrol Start
Fri Sep 9 20:57:05 GMI 1994: Patrol End
Fri Sep 9 20:57:57 GMI 1994: Fil emark
Mauna Loa Sol ar Observatory Cbserver’s Log



Fri Sep 9 21:08:32 GV 1994
Year: 94 Doy: 252
onserver: yasukawa

DPMON COMMENT: Fri Sep 9 21:10:22 GMI 1994

Still got I/Oerrors on tape wites. | think the new tape needed
to be initialized. Rebooted in any case after protecting today’s
logs. Starting up again

Fri Sep 9 21:11:57 GMI 1994: Patrol Start

DPMON COMMENT: Fri Sep 9 21:16:55 GVII 1994
Looks like tape drive is bbehaving again. Reconfigured back to
i nb-preferred node.

WEATHER COMMENT: Fri Sep 9 21:18:33 GMI 1994
Crrostratus burned off but now orographic clouds are starting
to form overhead.

COMMENT: Fri Sep 9 21:57:45 GMTI 1994

Automatic done rotator is on, the slot is quite tight this time of year
so the ADR is overconpensating a bit causing the MKIII to get

doned out briefly each tine the donme noves around ni dday.

Fri Sep 9 22:02:57 GMI 1994: Calibration

MKI 11 COMMENT: Fri Sep 9 22:29:20 GMI 1994
Manual |y noved donme to favor MKIII for a few scans.

DPMON COMMENT: Fri Sep 9 23:02:43 GMI 1994
Still struggling to keep the imge occulted. |Increasing the size
of the occulter one nore step, from 1941 to 1951

Fri Sep 9 23:10:06 GVI 1994: Patrol Start

WEATHER COMMVENT: Fri Sep 9 23:50:59 GMI 1994

In thin cirrus.

O ographi ¢ cl ouds novi ng upsl ope to west of station.

Fri Sep 9 23:58:03 GMI 1994: Patrol End
Sat Sep 10 00: 06: 31 GMI 1994: Patrol Start

WEATHER COMMENT: Sat Sep 10 00: 07: 10 GMI 1994
Clear again



WEATHER COMMENT: Sat Sep 10 00: 24:27 GMI 1994
Sun di pped into the orographic clouds noving upslope to west.

WEATHER COMMENT: Sat Sep 10 00:59:31 GV 1994
In thin cirrus, in and out of fast noving orographic clouds.
Sat Sep 10 01:00:56 GMI 1994: Cali bration

WEATHER COMMENT: Sat Sep 10 01:02:53 GMI 1994
Lotsa cirrus. Stopping.

Sat Sep 10 01:03:59 GMI 1994: Patrol End

COMMENT: Sat Sep 10 01:05:13 GVII 1994

Activity report:
QP:  90; 140; 226; 252; 290; 305;
No coronal activity

Tapes: MIII: HO0837
DPMON:  P00155 I/O error failure at end of data
P00156

LONML: LOO104 in unit 1

Sat Sep 10 01:11:51 GMI 1994: Fil emark

SCAN- LOG

SCAN- LOG 16:51: 36. 9/ 9/ 94 DOY 252
16: 58: 26 17:01: 40 17:04: 56 17:08: 09 18: 50: 42
18: 53: 51 18:57: 02 19: 00: 12 19: 03: 24 19: 06: 33
19: 09: 44 19:12: 53 19: 16: 02 19:19: 11 19: 22: 22
19: 25: 31 19: 28: 41 19: 31: 50 19: 35: 00 19: 38: 09
19: 41: 20 19: 44: 29 19: 47: 38 19: 50: 46 1954 0 CL
19:58: 15 2003 15CL 20: 07: 21 2011 0 CL 20: 14: 25
2018 15CL 20: 23: 22 20: 26: 54 20: 30: 04 20: 33: 15
20: 36: 26 20: 39: 36 20: 42: 46 20: 46: 00 20:49: 08
20:52: 17 20: 55: 26 20:58: 36 21:01: 44 21:04: 58
21: 08: 06 21:11: 14 21: 14: 22 21:17: 31 21: 20: 38
21: 23: 51 21: 27: 00 21: 30: 09 21:33: 17 21:36: 31

21:39: 39 21:42: 48 21:45: 56 21:49: 05 21:52:12
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